Author's response to reviews

Title: Familial tumoral calcinosis in two Chinese patients: a case series

Authors:

Che Zhang (prof.zh@163.com)
Jiaowei Gu (giw888giw@163.com)
Xiaoli Chen (cattych163@163.com)
Kui Xiong (xiongkui@cSCO.org.cn)

Version: 4 Date: 20 February 2011

Author's response to reviews: see over
Cover Letter

MS:  2094385716459480

Familial tumoral calcinosis in two Chinese patients: a case series
Che Zhang, Jiaowei Gu, Xiaoli Chen and Kui Xiong

Comments from Associate Editor:
1. "the references needs to be updated. there have been multiple papers and reviews in the last couple of years on tumoral calcinosis which have not been included in the paper."
Response: The references were updated within past 5 years except for the initial introduction of TC concept at reference No. 1. Besides, 2 papers in 2010 were updated in reference as No. 3 and 8.

2. “The discussion can be made more crisp.”
Response: The clinical classification and treatment in discussion section were simplified.